Author’s response to reviews

Title: A video-based transdiagnostic REBT universal prevention program for internalizing problems in adolescents: study protocol of a cluster randomized controlled trial

Authors:

Costina Păsărelu (costinapasarelu@gmail.com)
Anca Dobrean (anca.dobrean@ubbcluj.ro)

Version: 1 Date: 13 Sep 2017

Author’s response to reviews:

Dear Editor,

Please find attached the revised version of the manuscript BPSY-D-17-00548 entitled “Efficacy and mechanisms of change of video-based transdiagnostic REBT universal prevention program delivered in a school context for internalizing problems in adolescents: study protocol of a cluster randomized controlled trial”, authored by Costina R. Păsărelu and Anca Dobrean. All authors are in agreement with the contents of the manuscript.

In this manuscript we described the protocol of a video-based transdiagnostic universal prevention program that we will implement in Romanian schools. We consider that these findings could have important theoretical and practical implications, in designing mental health prevention and interventions programs for adolescents.

This study has not been published elsewhere and is not under consideration in any other journal. No conflicts of interest have to be disclosed for the authors.

I confirm that I have forwarded the requested documentation regarding ethical and funding to BMCSeriesEditorial@biomedcentral.com. I also confirm that the present study has received external funding and therefore, the study protocol has undergone peer-review by the funding body.

Thank you for your consideration.

Yours sincerely,

Anca Dobrean, Ph.D.
Department of Clinical Psychology and Psychotherapy
Babes-Bolyai University
37 Republicii St., 400015, Cluj-Napoca, Romania
Tel/Fax: +40264434141
Email: anca.dobrean@ubbcluj.ro